Author’s response to reviews

Title: Inequalities and factors associated with adherence to diabetes self-care practices amongst patients at two public hospitals in Gauteng, South Africa

Authors:

Chipo Mutyambizi (CMutyambizi@hsrc.ac.za; mutyambizic@gmail.com)

Milena Pavlova (m.pavlova@maastrichtuniversity.nl)

Charles Hongoro (chongoro@hsrc.ac.za)

Wim Groot (w.groot@maastrichtuniversity.nl)

Version: 2 Date: 09 Jan 2020

Author’s response to reviews:

We thank the reviewers for their useful comments and suggestions which greatly helped us to improve our paper. Below we explain how we have addressed each of the reviewers’ comments.

Editor Comments:

1. The corresponding author information as provided in the manuscript does not match the corresponding email address provided on the editorial manager system. Please amend this. Please also clearly state the corresponding author email address on the title page.

Response: The corresponding author information has been updated

2. Please move the abbreviations and declarations to above the references.

Response: The above changes have been made

3. In the ethics approval and consent to participate section, please clarify if written informed consent was obtained from all participants.

Response: Written informed consent was obtained from all participants. This has been stated in the ethics approval and consent to participate section.
4. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: The role of the funding body has been clarified in the funding section

5. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: This has been noted and the response to reviewers/editors’ comments have been inserted in the Response to Reviewers box in Editorial Manager